Author’s response to reviews

Title: Efficacy of Removing Candida albicans from Orthodontic Acrylic Bases: An in vitro study

Authors:
Abdul Razzak Ghazal (dr-ghazal@hotmail.com)
Ghassan Idris (ghassan.idris@otago.ac.nz)
Mohammad Hajeer (myhajeer@gmail.com)
Karam Alawer (drkaramalawer@gmail.com)
Richard Cannon (richard.cannon@otago.ac.nz)

Version: 4 Date: 11 Apr 2019

Author’s response to reviews:

Dear Prof. Messin, Editor of BMC Oral Health:

Please be informed that we accomplished what is requested from the reviewers as follows,

1- Page 11, line 1:
The concentration of CHX has been corrected to become 0.12% instead of 0.012%

2- We have reduced the number of references by deleting 4 references which are:
(8) Traore et al, 2002
(32) Chang et al, 2014
(33) Khanpayeh et al, 2014
(35) Nikawa et al, 2005

Thanks for the helpful comments,

Best regards,

Authors